Author's response to reviews

Title: Cross-cultural adaptation of the Neck Disability Index and Copenhagen Neck Functional Disability Scale for patients with neck pain due to degenerative and discopathic disorders. Psychometric properties of the Polish versions.

Authors:

Ewa Misterska (emisterska1@wp.pl)
Roman Jankowski (klinikanch@op.pl)
Maciej Glowacki (glowackimaciej@o2.pl)

Version: 4 Date: 14 April 2011

Author's response to reviews: see over
Authors' response to the Editor’s remarks

Version 3

Title: Cross-cultural adaptation of the Neck Disability Index and Copenhagen Neck Functional Disability Scale for patients with neck pain due to degenerative and discopathic disorders. Psychometric properties of the Polish versions.

Authors: Ewa Misterska, Roman Jankowski, Maciej Glowacki

Formatting changes to be made:

-----------------------------

Major revisions (we require the author to make these changes)

Abstract: This should be identical in the manuscript and the submission system. Please ensure that the Trial Registration section is either removed, or included in both versions.

We will make sure abstract will be identical in the manuscript and the submission system and the Trial Registration section will be either removed, or included in both versions.

References: Please make sure references are cited in numerical order.

The References are cited in numerical order now.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Abstract: please remove the Keywords section from your abstract, as follows:

We have removed this section from our abstract.

Authors' contributions: please ensure this is formatted into a paragraph rather than a list or in bullet points.

We have formatted authors' contributions into a paragraph.
Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

*We have removed all highlighting and tracking from our manuscript.*

Tables- please remove the visible vertical lines from your tables

*We have removed all visible vertical lines from our tables.*

Tables: please note that we are unable to correctly display merged cells where the merged cell crosses rows: please re-layout your table without these merged elements

*We have corrected the Tables and removed the merged cells. We hope the Tables are now suitable.*

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

*We have checked our manuscript for any typographical errors.*